Author’s response to reviews

Title: Gender differences in the prevalence of congenital heart disease in Down's syndrome: a brief meta-analysis.

Authors:

Tereza Diogenes (tereza.diogenes@hotmail.com)
Felipe Mourato (felipe.a.mourato@gmail.com)
José Lima Filho (joseluiz60@mac.com)
Sandra Mattos (ssmattos@gmail.com)

Version: 2 Date: 13 Sep 2017

Author’s response to reviews:

To:

Dr. Matteo Pasini,

Dear Dr. Pasini:

We are submitting the revised version of the manuscript entitled “Gender difference in the prevalence of congenital heart disease in Down’s syndrome: a brief meta-analysis.” by Diogenes et al as a Research Article for publication in BMC Medical Genetics. The manuscript has been revised and altered to deal with your concerns.

All authors have read and approved the revised version of the manuscript; the manuscript has not been published and is not being considered for publication elsewhere, in whole or in part, in any language, except as an abstract. In addition, the authors declare no conflict of interest.

We thank you in advance and hope this manuscript is now suitable for publication in the BMC Medical Genetics.

Sincerely yours,

Felipe Alves Mourato